Author's response to reviews

Title: Cardiac condition during cooling and rewarming periods of therapeutic hypothermia after cardiopulmonary resuscitation

Authors:

Serdar Demirgan (serdardemirgan@hotmail.com)
Kerem Erkalp (keremerkalp@hotmail.com)
M. Salih Sevdi (salihseyd@gmail.com)
Meltem Turkay Aydogmus (meltem72_3@hotmail.com)
Numan Kutbay (nkutbay@hotmail.com)
Aydin Firincioglu (aydinfirincioglu@gmail.com)
Ali Ozalp (ali_ozalp3@hotmail.com)
Aysin Alagol (aysinalagol@yahoo.com)

Version: 5 Date: 27 June 2014

Author's response to reviews: see over
Dear Editor;

Our manuscript serial number is “2054454991223597” and its title is “Cardiac condition during cooling and rewarming periods of therapeutic hypothermia after cardiopulmonary resuscitation”… Authors: Serdar Demirgan, Kerem Erkalp, M. Salih Sevdi, Meltem Turkay Aydogmus, Numan Kutbay, Aydin Firincioglu, Ali Ozalp and Aysin Alagol

We thank you for your polite response. We are very happy from your interest to our manuscript.

The manuscript was revised and the comments were added point-by-point responses.

As you want, the demographic table (Table 1) was kept. The remaining tables were removed. Table 12 was restructured to include each parameter, side by side listing of the mean +/- SD at each body temperature during cooling vs rewarming, and the p value in an adjacent column. And, it was renamed Table 2.

Mistyped of reference 33 and 41 was corrected.

Finally, our manuscript was re-edited by Edanz Language Editing Service.

We hope that this last form of our manuscript is suitable for your readers.

Sincerely yours.

26.06.2014

Dr. Kerem Erkalp

On behalf of all the authors.